## Shareholder Power and the Decline of Labor\*

Antonio Falato<sup>†</sup>

Hyunseob Kim<sup>‡</sup>

Till von Wachter<sup>§</sup>

February 2021

#### Abstract

Using confidential establishment-level data from the U.S. Census Bureau's Longitudinal Business Database from 1982-2015, we examine how shareholder power affects firm employment and payroll decisions. Consistent with theory of the firm based on conflicts of interests between shareholders and stakeholders, we find that establishments of firms owned by larger and more concentrated institutional shareholders have lower employment and payroll. These results are not driven by unobserved heterogeneity across establishments or differences in exposure to industry and local shocks, and hold up in a difference-in-differences design that exploits large increases in block institutional ownership. The results are more pronounced in industries with a smaller fraction of unionized labor, in more concentrated local labor markets, and for dedicated and activist institutions. The labor losses are accompanied by higher shareholder returns but lower labor productivity. Our findings suggest a role for employment policies that aim at reforming shareholder capitalism.

Key words: Shareholder power; institutional ownership; concentration; employment; wages

<sup>\*</sup> We thank Philip Bond, Murillo Campello, Luke Stein, Amir Sufi, and seminar participants at Australian National University, Babson College, and Cornell University for helpful comments, and Vitor Costa and Katya Potemkina for excellent research assistance. Any views expressed are those of the authors and not those of the U.S. Census Bureau. The Census Bureau's Disclosure Review Board and Disclosure Avoidance Officers have reviewed this information product for unauthorized disclosure of confidential information and have approved the disclosure avoidance practices applied to this release. This research was performed at a Federal Statistical Research Data Center under FSRDC Project Number 1572. (CBDRB-FY20-P1572-R8820) We thank Charles Hokayem and Nichole Szembrot for helping with data and clearance requests. All errors are our own.

<sup>&</sup>lt;sup>+</sup> Federal Reserve Board; Email: antonio.falato@frb.gov; Phone: (212) 662-1185.

<sup>&</sup>lt;sup>‡</sup> Johnson School of Management, Cornell University; Email: hk722@cornell.edu; Phone: (607) 255-8335.

<sup>&</sup>lt;sup>§</sup> Department of Economics, UCLA; Email: tvwachter@econ.ucla.edu; Phone: (310) 825-5665.

### 1. Introduction

The decline of labor, amid stagnant wages and falling employment, is a well-documented stylized fact of US labor markets over the last decades (see, e.g., Pierce and Schott, 2016; and Autor, Dorn, Katz, Patterson, and Van Reenen, 2020). At the same time, public corporations, which account for a large share of private employment, underwent radical changes in their ownership structure with the rising importance of large institutional shareholders. Figure 1 plots the average block institutional ownership (defined as more than 5% holdings) shares for US public firms and the ratio between aggregate wages and salaries to domestic gross income (GDI) in the US from 1980 to 2014. It shows that the measure of large institutional ownership increased almost threefold over the period, while the "labor share" of domestic income exhibited a steady decline. These concurrent trends suggest that there may be a link between the ownership structure of public corporations and their employment decisions. In fact, prominent commentators in the popular press and even CEOs of large corporations have expressed concerns that the so-called "shareholder capitalism" – i.e., the North American style of governance that since the 1980s has centered around the maximization of shareholder value as the main objective of the firm – may have hurt workers.<sup>1</sup> Yet, there is a dearth of systematic evidence on the role of large institutional shareholders in the governance of labor.

In an attempt to fill the gap, this paper uses confidential micro data from the U.S. Census Bureau's Longitudinal Business Database (LBD). Specifically, we assemble a rich establishment-level dataset that spans the 1982 to 2015 period and contains annual information on employment, payroll, and ownership structure for over seven million establishment-year observations. Using this dataset, we show a robust negative within-establishment relation between shareholder power, measured by ownership of large institutional shareholders, and firm employment and payroll.

Our analysis is guided by the classical agency theory of the firm, which dates back to Jensen and Meckling (1976) (see Shleifer and Vishny, 1997; and Stein, 2003, for comprehensive reviews). In this class of theory, there is a fundamental conflict of interests over the allocation

<sup>&</sup>lt;sup>1</sup> See, for example, Business Roundtable Statement on the Purpose of the Corporation (2019) at https://opportunity.businessroundtable.org/ourcommitment/, Posner (2019), Krugman (2015), and Stiglitz (2019).

of firm resources between shareholders and stakeholders, including workers. Shareholders with large and concentrated ownership could hurt workers through two distinct but related channels: first, they can more easily renege ex post on implicit contracts not to fire workers or cut payroll via hostile takeovers or restructurings, as in the "breach of trust" hypothesis of Shleifer and Summers (1988); second, they can more easily monitor managers and force them to fire workers or cut payroll against their will, as per the "quiet life" hypothesis of Bertrand and Mullainathan (2003). In both cases, shareholder power hurts labor.



Figure 1: Trends in Average Block Institutional Ownership of Public Firms and Employee Compensation-to-GDI Ratio in the US, 1980–2014. This figure plots the average block institutional ownership for public firms (blue solid line; left panel) and the ratio of total wage and salary accruals to domestic gross income (GDI) (red dotted line; right panel) in the US for the period 1980–2014. The red lines in both panels show linear time trends. Institutional ownership is from Thomson Reuter 13F filings data and the compensation-to-GDI ratio is from the Federal Reserve Economic Data. Block institutional ownership is the percentage owned by institutional blockholders, defined as the institutional investors with more than 5% holdings as filed through 13D, 13F, or 13G filings. The average block ownership is calculated excluding firms in the agriculture, financial, utility, and public administration sectors.

Consistent with this agency view of the firm, we find that establishments of firms owned by powerful institutional shareholders have significantly lower employment and payroll within establishments (i.e., using establishment fixed effects). The relation holds robustly for several proxies for shareholder power, including the percentage ownership of the largest one or five institutional shareholders, blockholders (defined as the institutional shareholders with at least a 5% stake), and the Herfindahl-Hirschman Index (HHI) of institutional investors' ownership. Across the measures of shareholder power, the relation is economically significant: a 10-percentage point increase in large shareholder ownership is associated with up to a 2.5% reduction in a given establishment's employment and payroll. The negative association between shareholder power and employment and payroll holds up to several batteries of specification and robustness checks, which include controlling for other confounding influences due to time-varying industry and local market conditions, as well as limiting the sample to just manufacturing establishments and using symmetric employment or payroll growth as the outcome variable. Finally, we confirm the negative relation in a difference-indifferences (DD) design that exploits large changes in powerful institutional ownership to addresses selection issues.

After establishing the negative relation, we take a first step toward clarifying the mechanism by determining which powerful shareholders have the larger impact on labor and when. First, we explore variation in bargaining power. The negative relation varies predictably with labor bargaining power, and the employment and payroll losses are larger in sectors where a smaller fraction of workers is unionized and in local labor markets where firms have more bargaining power vis-à-vis workers. Second, the labor losses are accompanied by higher shareholder returns but lower labor productivity. The loss of output and efficiency is hard to reconcile with a monitoring channel and more in line with a wealth transfer from workers to shareholders. Finally, we find that only institutional shareholders with control motives -i.e., "dedicated" institutions according to Bushee's (1998) classification or "activist" institutions have a strong negative association with employment and payroll. Other types of noncontrolling shareholders, such as "quasi-index" or "transient" institutions, have a weaker and if anything, positive relation with employment and payroll. Anecdotal evidence from shareholder activism campaigns further points to the maximization of shareholder value as the stated goal to manage labor-related operating expenses. In all, the collection of evidence indicates that shareholder power has mainly a reallocative impact, as labor cuts take value away from workers and toward shareholders without increasing total firm revenues.

While there is an active literature on the determinants of the decline of labor, the role of corporate governance and ownership has received limited attention. There are a number of explanations that have been set forth in the literature, including technology (Acemoglu, 2002), import penetration (Autor, Dorn, and Hanson, 2013; Acemoglu, Autor, Dorn, Hanson, and Price, 2016; and Pierce and Schott, 2016), industry concentration and superstar firms (Autor, Dorn, Katz, Patterson, and Van Reenen, 2020), and labor market concentration (Benmelech, Bergman, and Kim, 2020). A related recent literature also examines the role of firms in wagesetting (see, for example, Card, Heining, and Kline, 2013; and Song, Price, Guvenen, Bloom, and von Wachter, 2019). Our findings are complementary to existing explanations, and suggest a role for employment policies aimed at reforming shareholder capitalism and broadening the objective of the firm beyond the narrow maximization of shareholder value, such as increasing labor representation on boards or allowing for labor participation in firm governance.

We also contribute to the literature on corporate governance and labor by exploring the link between corporate ownership structure and its labor decisions. This literature shows that measures of managerial entrenchment, such as state-level anti-takeover statues and dualclass shares, are associated with higher employee wages (Bertrand and Mullainathan, 2003; and Cronqvist et al., 2009). Another strand of this literature examines the impact on labor of specific types of institutions, such as private equities (Lichtenberg and Siegel, 1990; and Davis et al., 2014) and activist hedge funds (Brav, Jiang, and Kim, 2015). Our contribution is to provide to the best of our knowledge the first evidence on the labor impact of shareholder power for the universe of large institutional shareholders. Our findings indicate that, while there is interesting heterogeneity depending on institution type, concentrated institutional ownership matters for labor across a broad spectrum of institutions and in a representative sample of establishments owned by US public corporations.

The rest of this paper is organized as follows. Section 2 describes the data and variables used, and provides descriptive statistics. Section 3 details the main empirical specification, our main establishment-level results, and covers several specification and robustness checks. Section 4 explores mechanisms underlying the main results. Section 5 concludes.

#### 2. Data and descriptive statistics

This section describes the data sets used in the empirical analysis, sample selection procedures, and resulting samples.

#### 2.1. Data sources and sample construction

Our main data source for establishment observations is the Longitudinal Business Database (LBD) from the US Census Bureau. The LBD is a comprehensive dataset of establishments in the United States, covering all private business establishments with at least one employee (Jarmin and Miranda, 2002). The dataset provides annual establishment-level information on employment, payroll, industry classification, and geographic location (e.g., counties and states) as of March 12.<sup>2</sup> We use the log of establishment-level employment and inflation-adjusted payroll (in 2005 constant dollars) as main outcome variables in our empirical analysis.

In addition, we use a set of control variables standard among research analyzing establishment-level data employing Census micro datasets (see, for example, Schoar 2002; and Benmelech, Bergman, and Kim, 2020). Specifically, firm size is measured by the log number of establishments of a given firm, while firm segment size, defined at the firm-industry level, is measured by the log number of establishments belonging to the firm in a given three-digit standard industry code (SIC) industry. Establishment age is defined as the number of years since an establishment's inception—identified by the flag for establishment's inception in the LBD—or its first appearance in the LBD, whichever is the earliest. The starting year is censored in 1977 when the coverage of the LBD begins.

The institutional ownership data are from Thomson Reuters 13F SEC filings. All institutional investment managers with greater than \$100 million in equity assets under discretionary management are required to file a Form 13F with the Securities and Exchange Commission on a quarterly basis. All common stock holdings of 10,000 or more shares or having a value of \$200,000 or more must be reported. Qualified securities include stocks listed for trading in the U.S., among other securities. The quarterly holdings reported in Form 13F represent the aggregate holdings of an institution (e.g., the Vanguard family of funds), rather than the holdings of any individual portfolio (e.g., the Contra fund in the Fidelity family of funds). Throughout the paper, an institutional investor (or shareholder) is defined as an institution that files a 13F.

<sup>&</sup>lt;sup>2</sup> The LBD employment includes full-time and part-time workers.

The main ownership variables of interest are defined as the holdings in a stock by a given type of institutional investor as a percentage of shares outstanding. "Overall IO" is the percentage owned by all institutional investors. "Top 1 ownership" and "Top 5 ownership" are the percentages owned by the largest one and five institutional investors, respectively. "Total block ownership" is the percentage owned by all blockholders, which are defined as the institutional investors with more than 5% holdings as filed through 13D, 13F, or 13G. To better distinguish between ownership types among institutional shareholders, we also consider three additional variables: "Quasi-index IO," "Dedicated IO," and "Transient IO," which are the percentages owned by quasi-index, dedicated, and transient institutional investors as classified by Bushee (1998), all scaled by "Overall IO." "Activist IO" is the percentage owned by activist institutional investors as classified by Grennan (2019) based on institutional investors that have engaged in shareholder activism campaigns, scaled by "Overall IO."

We require that each establishment observation from the LBD have non-missing, positive values for employment and payroll, as well as the one-year lagged observation, which is used in part of our analysis that uses growth rates of employment or payroll as an outcome. We then match establishments with firm-level institutional ownership variables from Thomson Reuters as of the first quarter of the previous year using the Census Bureau's bridge file between the firm identifiers in the LBD and Compustat.<sup>3</sup> We require that establishments have the matched institutional ownership variables. We exclude the agriculture (SIC 0100-0900), utilities (SIC 4900-4999), financials (SIC 6000-6999), and public administration (SIC 9000-9999) industries from our sample. This sample selection procedure yields approximately 7,340,000 establishment-year observations from 1982–2015 (the Thomson Reuters data are available from 1981–2014). A subsample for the manufacturing sector (SIC 2000-3999), which is used in part of our analysis, includes approximately 533,000 establishment-year observations. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

<sup>&</sup>lt;sup>3</sup> LBD information is as of March of a given year. Thus, institutional ownership as of the first quarter of the previous year represents the ownership information that is lagged by a full year.

We construct data on collective bargaining coverage by following the approach in Hirsch and Macpherson (2003).<sup>4</sup> Below we provide a detailed account of the data source and construction that are relevant for our empirical analysis. We compute the fraction of workers covered by labor unions at the industry level using the Current Population Survey (CPS) Outgoing Rotation Groups (ORG) data.<sup>5</sup> The resulting dataset provides union coverage rate estimates, defined as the number of employees covered by labor unions divided by total employment, by Census Industry Code (CIC) beginning in 1983. Because our establishmentlevel data use SIC codes, we match the union coverage data with Census establishment-level data at the industry level as follows.<sup>6</sup> For the years 1983–2002, we use the 1980 and 1990 Census Bureau's concordances between CIC codes and SIC codes. For 2003–2015, during which the CIC is based on NAICS codes, we first match 2000 CIC codes to 1990 CIC codes using the Census Bureau's concordance and then use the Census Bureau's concordance between 1990 CIC and SIC codes to match with the Census establishment-level data.<sup>7</sup> Given that our industry-level database on union coverage begins in 1983 whereas our sample period starts in 1982, we impute the rate of collective bargaining coverage using 1983 information for years 1981 and 1982.8

#### 2.2. Descriptive statistics

Table 1 shows descriptive statistics for institutional ownership and establishment characteristics for establishments in the full sample (Panel A) and in the manufacturing sample (Panel B) from 1982-2015. Focusing on the full sample, the average overall institutional ownership is 59.7%, consistent with rising institutionalization of public firm ownership in the past decades. The average ownership by blockholders, defined as institutional investors with

<sup>&</sup>lt;sup>4</sup> For the data used in Hirsch and Macpherson (2003) see <u>www.unionstats.com</u>.

<sup>&</sup>lt;sup>5</sup> The database is available at <u>http://www.nber.org/data/morg.html</u>.

<sup>&</sup>lt;sup>6</sup> The CPS ORG data employs the 1980 CIC classification (based on SIC codes) from 1983 to 1991, the 1990 CIC classification from 1992 to 2002 (based on SIC codes), and the 2000 CIC (based on NAICS codes) from 2000 and on.

<sup>&</sup>lt;sup>7</sup> For the majority of CIC industries used in the CPS, this matching procedure results in direct linkages to threedigit SIC industries. In a minority of cases, the procedure results in a match to a two- or four-digit SIC industry, in which case the finer industry classification is used.

<sup>&</sup>lt;sup>8</sup> The Census establishment-level datasets use different vintages of SIC and NAICS codes across years (see for example, Table 1 in Fort and Klimek, 2018). Thus, we use the appropriate vintage of the industry codes for each year (for example, SIC1997 for 1997).

at least 5% ownership, is 14.7% and exhibits considerable variation with a standard deviation of 14.5%. Ownership by "dedicated" institutions, which are likely active monitors, accounts for 13.1% of all institutional ownerships. These statistics suggest that a bulk of institutional shareholding is across dispersed, less active owners.

#### [Insert Table 1 here.]

Turning to establishment characteristics, the average establishment in the full sample employs about 62 workers and pays about \$29,500 per worker (in 2005 constant dollars). The (public) parent firm of the average establishment owns more than 2,000 establishments, and the average age of establishments is 10.8 years. Panel B shows similar mean values of institutional ownership variables for manufacturing establishments. They however exhibit larger employment and payroll on average, relative to the establishments in the full sample. The average annual pay is about \$42,700 per worker.

#### 3. Empirical analysis

This section provides baseline estimates for the within-establishment relation between large shareholder ownership and employment or payroll, explores large increases in institutional ownership as a quasi-natural experiment, and examines robustness.

### 3.1. Baseline results – ownership by top institutional shareholders, blockholders, and ownership concentration

We estimate the effect of ownership by large institutional shareholders, which presumably increases shareholder power, on establishment-level employment and payroll using the following regression:

$$y_{ijkt} = \alpha_i + \alpha_{kt} + \beta_1 Large IO_{jt-1} + \beta_2 Overall IO_{jt-1} + \gamma' X_{it-1} + \varepsilon_{ijkt},$$
(1)

where  $y_{ijkt}$  is either of the log of total employment or payroll, *i* indexes establishment, *j* firm, *k* industry, and *t* year;  $a_i$  is establishment fixed effects;  $a_{kt}$  is industry-by-year fixed effects; *Large*  $IO_{jt-1}$  is a measure of large institutional shareholders' ownership (see below); *Overall IO<sub>jt-1</sub>* is overall institutional ownership;  $X_{it-1}$  is a set of establishment-level control variables, including firm size and firm-segment size, and  $\varepsilon_{ijkt}$  is the residual. Standard errors are clustered at the firm level.

The coefficient of interest is  $\beta_1$ , which represents the within-establishment semielasticity of establishment-level employment or payroll to a measure of large institutional ownership, after controlling for overall institutional ownership, establishment-level covariates, and time-varying industry conditions. We first employ ownership by top institutional investors as a measure of large shareholder ownership, and then employ total institutional blockholder ownership and concentration of ownership across institutions. The premise is that institutional investors would have strong incentives to maximize shareholder value and power to do so, potentially at the expense of employees, when they have large fractional ownership (Shleifer and Vishny, 1986) or when ownership is concentrated in the hands of few institutional investors. The control for overall institutional ownership (represented by the coefficient  $\beta_2$ ) helps address alternative interpretations due to omitted factors not related to institutions' incentives to monitor the firm. If our main estimates were due to omitted factors that drive institutions' decisions to buy or sell shares in a given firm for reasons unrelated to monitoring, then we would expect our *Large IO* variables have no incremental explanatory power after controlling for overall institutional ownership.

Table 2, Panel A shows estimation results for Eq. (1) using the top one and five institutional ownership variables on the full industry sample. Columns 1 and 2 (columns 3 and 4) present the baseline within-establishment relation between ownership of the largest institutional shareholder and log employment (payroll). The coefficient on "Top 1 ownership" in column 1 is -0.217 and significant at a less than 1% level. This estimate shows that for a given business establishment, a ten-percentage point increase in the largest institution's ownership (relative to the mean of 8.7%) is associated with a 2.2% decline in employment, controlling for time-varying industry conditions and establishment decline in payroll (by 2.1%, *t*-stat = 5.22) in response to a ten-percentage point increase in the largest institution's ownership. These significantly negative relations are consistent with the idea that when the firm's largest institutional shareholder increases ownership and thus incentives to monitor (Shleifer and Vishny, 1986) or "power," it reduces employment and payroll at the establishments in an attempt to maximize shareholder value.

Similarly, columns 2 and 4 show that the collective ownership of top five institutional shareholders is negatively associated with establishment-level employment and payroll (significant at a less than 1% level). A ten-percentage point increase in top five ownership is associated with 2.3% and 2.5% declines in number of employees and wage bills, indicating a similar magnitude of the association with that for top one ownership.

Across the columns, the coefficients on control variables have expected signs – the log number of establishments per firm, a measure of the owner firm's scope, is significantly positively correlated with both establishment-level employment and payroll. Importantly, the coefficient on the overall institutional ownership measure is positive and generally smaller in magnitude relative to that of our main variables of interest. The positive relation between general institutional ownership and employment and payroll indicates that, unless they are powerful, the presence of institutional shareholders does not in and of itself harm workers. This can be the case either because diffused institutional shareholders do not have the ability to influence firm decisions on labor or because their objectives (or incentives) differ from those of large institutional shareholders, as it may be the case for passive buy-and-hold investors. We explore this possibility in more detail in the analysis by institution type in the next section.

### [Insert Table 2 here.]

Panel B presents estimates of Eq. (1) for a subsample of manufacturing establishments. Similar to the full sample results in Panel A, the coefficients on top one and five ownership variables are consistently negative and significant at a less than 1% level. The economic magnitude of the effect is somewhat larger in manufacturing than across all industries – for example, estimates in columns 2 and 4 indicate that a ten-percentage point increase in top five institutional ownership is associated with 2.9% and 3.3% within-establishment reductions in employment and payroll. This larger magnitude of decline in employment and payroll in manufacturing is consistent with existing work documenting that the decline of labor is most pronounced in manufacturing (e.g., Autor et al., 2020).

Next, we explore how powerful institutional shareholders affect establishment-level employment and payroll by estimating Eq. (1) using total institutional blockholder ownership

and the Herfindahl-Hirschman Index (HHI) of institutional investors' ownership as measures of shareholder power.

## [Insert Table 3 here.]

Table 3 shows estimation results using the total institutional blockholder ownership and HHI variables. The coefficients on "Total block ownership" in columns 1 and 3 in Panel A show that, when a given establishment experiences a ten-percentage point increase in block institutional ownership, its employment and payroll decline by 1.5% and 1.7%. The coefficients are significant at a less than 1% level. Similarly, the significantly negative coefficients on "HHI (institutional ownership)" in columns 2 and 4 show that when institutional ownership becomes more concentrated, the establishment exhibits smaller employment and wage bills. A one-standard deviation (SD) increase in the HHI (0.110) would lead to an about 1% reduction in establishment employment and payroll within establishments.

Panel B presents estimates for establishments in the manufacturing sector, showing similarly negative within-establishment relations between changes in block ownership or concentration of ownership and employment and payroll. The economic magnitude of the effect is particularly more pronounced in manufacturing when the HHI is employed to measure shareholder power – the coefficient on the HHI is 52% (employment) to 104% (payroll) larger in manufacturing relative to the full set of industries as shown in Panel A.

#### 3.2. Robustness

A potential concern for our baseline results is that ownership of large institutional investors is correlated with local labor market conditions, which would affect labor market outcomes such as employment and wages. For example, spurious correlation may arise if large institutional investors target firms operating in smaller labor markets, in which employment and wages tend to be lower (e.g., Moretti, 2011). To mitigate this type of concern, we reestimate the baseline regression in Eq. (1) with county-by-year fixed effects, and present estimation results in Panel A.1 of Table 4.

#### [Insert Table 4 here.]

Across the columns, the panel shows that the estimated association of top one or five, block, or concentrated institutional ownership with labor quantities is generally more pronounced once time-varying local labor market conditions are controlled for. For example, relative to the comparable estimate in Table 2, Panel A.1, column 1 (-0.217), the coefficient on "Top 1 ownership" in Table 4, Panel A.1, column 1 (-0.286) indicates an employment effect that is about one-third larger. Similarly, the estimated semi-elasticity of payroll to top one ownership in column 5 is about one-third larger than the comparable estimate in Table 2, Panel A, column 3 (-0.274 vs. -0.205). Overall, the results in Table 4, Panel A.1 show that omitted time-varying local market conditions are unlikely to explain the baseline results.<sup>9</sup>

We also check for robustness of the baseline results to using a moving average of institutional ownership, which includes two recent lags, to allow for potential dynamic effects due to a delayed response of employment and payroll to ownership. Estimates in Table 4, Panel B show that the effect of powerful institutional ownership, measured using moving averaged ownership, is consistently negative and highly statistically significant (largely at a less than 1% level). Interestingly, the economic magnitude of the effect is generally larger than those of the baseline results in Tables 2 and 3, indicating that further lagged changes in large institutional ownership. For example, the coefficient on "Top 5 IO (MA)" in column 2 (-0.321) is almost 40% larger than that on "Top 5 IO" in column 2 in Table 2, Panel A (-0.233).

In our main empirical analysis, we control for overall institutional ownership. Including this control is important to ensure that the main estimates capture the effect of shareholder power due to large and concentrated ownership by institutional shareholders. Intuitively, overall institutional ownership may not necessarily capture shareholder power, as firms with high levels of institutional ownership may still be widely held by many institutions, each with a small ownership share and thus limited power and incentives to monitor.

We examine the robustness of our baseline results by employing an alternative specification that scales the measures of top and blockholder ownership by overall institutional ownership, instead of including it as a control variable. This alternative measure gauges the relative importance of large or powerful institutional ownership among the overall institutional ownership (similar to the institution type shares in Section 4.3). The estimation results shown

<sup>&</sup>lt;sup>9</sup> Panel A.2 of Table 4 shows that the results are also robust to controlling for finer measures of local labor market conditions using industry-commuting zone-year fixed effects.

in Panel C of Table 4 are largely consistent with our baseline results in Tables 2 and 3. As the fraction of shares owned by top one or five institutions, or institutional blockholders among all institutional ownership increases, the employment and payroll of establishments decrease significantly (all coefficients are significant at the 1% level). For example, the coefficients on "Top 5 IO/overall IO" in columns 2 (-0.134) and 5 (-0.142) show that when the relative importance of top five institutional ownership among all institutional ownership increases by 10 percentage points, the establishment's employment and payroll would decline by 1.3% and 1.4%. Thus, the results in Panel C of Table 4 confirm that the negative within-establishment relation between large institutional ownership and employee outcomes is robust to how we control for overall institutional ownership.<sup>10</sup>

In additional robustness analysis, we employ an alternative specification that uses growth rates of establishment-level employment and payroll, instead of their log's, as dependent variables. Specifically, we estimate the following regression:

 $growth_{ijkt} = \alpha_{kt} + \beta_1 Large IO_{jt-1} + \beta_2 Overall IO_{jt-1} + \gamma' X_{it-1} + \varepsilon_{ijkt}$ , (2) where  $growth_{ijkt}$  is the symmetric growth rate of total employment or payroll, defined as  $(y_{ijkt} - y_{ijkt-1})/(y_{ijkt} + y_{ijkt-1})/2$ , *i* indexes establishment, *j* firm, *k* industry, and *t* year; and all other variables are defined as in Eq. (1), except that  $X_{it-1}$  also includes establishment age. We use a symmetric measure of employment and payroll growth following Davis, Haltiwanger, and Schuh (1996) to ensure that the result is not driven by the very largest employment or payroll increases and cuts. We do not include establishment fixed effects in Eq. (2), given that calculating the growth rates within establishments is akin to differencing out the level in the outcome variables.

Appendix Table A.1, Panel A shows estimation results of Eq. (2). The coefficients on top and block institutional ownership variables show a negative within-establishment relation between shareholder power and employment and pay growth rates, consistent with our baseline results for level changes. Interestingly, the negative relation is more pronounced for the growth rate of payroll relative to that of employment. For example, estimates in columns 1 and 5 show that a ten-percentage point increase in top institutional ownership is associated with a 0.3-percentage point slowdown in employment growth rate but with a 0.8-percentage

<sup>&</sup>lt;sup>10</sup> Panel D of Table 4 shows that the results are also robust to dropping the control for overall IO from the baseline specification.

point slowdown in payroll growth rate, more than a twice bigger magnitude. This finding indirectly suggests that greater ownership by large institutional investors is associated with lower per-employee wage growth. Columns 2-3 and 6-7 show that top five and total block institutional ownership variables are also associated with slower employment and wage growth, while columns 4 and 8 indicate a muted effect of the HHI of institutional ownership on the growth of labor. Overall, we find qualitatively consistent results using growth rates of the employee outcomes as dependent variable in place of the levels along with establishment fixed effects.

A final important robustness question relevant for macro implications of our results is how powerful institutional investors "allocate" cuts in employment and wage bills across establishments of different sizes. One possibility is that institutions would prefer cutting both employment and payroll at relatively large establishments within firms in an attempt to cut overall labor costs, which could benefit shareholders. This would imply a considerable aggregate effect of increased power of institutional investors both on employment and wages. Alternatively, powerful institutions with strong incentives to monitor may want to keep employment of larger establishments (which are likely more productive) but to lower their wages exploiting their bargaining power in an attempt to maximize shareholder value. This alternative scenario would imply that the negative effect of powerful shareholders on aggregate labor may be concentrated in wages rather than employment. We explore these possibilities by reestimating Eq. (2) with establishment observations weighted by their log employment.<sup>11</sup>

Panel B of Appendix Table A.1 shows estimation results. We find that when establishments are weighted by their size, the estimated relation between ownership of large institutions and employment growth is statistically insignificant and economically close to zero (columns 1 through 4). In contrast, columns 5 through 8 show that the presence of powerful institutional shareholders with strong incentives to monitor is associated with a significant reduction in establishment-level payroll growth. For example, the coefficient on "Top 5 IO" in column 6 is -0.073 and significant at a less than 1% level, comparable with an unweighted estimate in column 6 in Panel A (-0.079). A simple back-of-the-envelope calculation using

<sup>&</sup>lt;sup>11</sup> We prefer estimating Eq. (3) instead of Eq. (1) in weighting observations by establishment employment because (log) employment itself is one of the outcome variables in Eq. (1).

estimates in columns 2 (employment) and 6 (payroll) suggests that, when top five institutional ownership increases by 10 percentage points, the growth rate of per-employee wages would decrease by 0.56% (= (-0.073 – (-0.017)) × 0.10).

These results, combined with the baseline unweighted results, describe how large institutional investors cut employment and wage bills across establishments. For larger establishments (which are likely more productive hence require smaller cuts in employment), they would push for a cut in per-employee wages but not overall employment. For smaller establishments (which are likely less productive), they appear to push for reductions both in head counts and wages, representing an overall reduction in the scale of operations. Importantly, the employment-weighted estimates in Panel B suggest that the aggregate impact of large institutions' monitoring could be more on pay rather than employment, which is consistent with stagnation of wages over the past couple of decades (e.g., Acemoglu and Autor, 2011) along with the rise of institutional block ownership.

## 3.3. Identification – DD analysis of Large Increases in Institutional Ownership

One concern with our empirical approach so far is that powerful institutional shareholders may "cherry pick" firms they invest in. In other words, it is not the powerful institutional shareholders that lead to employment cuts, but rather that powerful institutional shareholders tend to invest in firms with declining labor. To address this identification challenge, which is an instance of a selection (or reverse causality) concern, we repeat our baseline analysis in a difference-in-differences (DD) setting that exploits events of "large" increases in institutional ownership. Specifically, we examine a subsample of establishments that experience a more than 5% increase in block institutional ownership as our "treatment" group. We then assemble a "control" group of establishments that experience a more than 5% increase in overall institutional ownership.<sup>12</sup>

Note that the identification assumption for this test is not that establishments experiencing a large change in block ownership are the same as those that experience a large change in overall institutional ownership. Rather, the identifying assumption is that there is no

<sup>&</sup>lt;sup>12</sup> Specifically, we require that the change in block ownership for the control establishments is less than 2%.

differential trend between the two groups pre-event. Using these samples, we estimate the following difference-in-differences equation:

$$y_{ijkt} = \alpha_i + \alpha_{kt} + \sum_{\tau=-3}^{-2} \lambda_\tau d[t+\tau]_{jt} + \sum_{\tau=0}^{4} \lambda_\tau d[t+\tau]_{jt} + \sum_{\tau=-3}^{-2} \delta_\tau d[t+\tau]_{jt} \times Treat_j + \sum_{\tau=0}^{4} \delta_\tau d[t+\tau]_{jt} \times Treat_j + \gamma' X_{it-1} + \varepsilon_{ijkt},$$
(3)

where  $d[t+\tau]_{jt}$  is an indicator variable equal to one if year t is  $|\tau|$  years before or after a large increase in block or overall institutional ownership, and zero otherwise ( $-3 \le \tau \le 4$ ). "Year t-1" is the baseline year and thus d[t-1] is equal to zero by construction *Treat<sub>j</sub>* is an indicator variable equal to one if establishments of firm *j* experience a more than 5% increase in block ownership, and zero if a more than 5% increase in overall institutional ownership. All other variables are defined as in Eq. (1).

In Panel A of Table 5, we first examine the characteristics of establishments that experience a large increase in block institutional ownership one year before the event relative to those that experience a large increase in overall institutional ownership. The results indicate that the two groups are comparable in terms of observable establishment characteristics such as employment size, age, and industry union coverage. Thus, there is no evidence of selection on these observables.

### [Insert Table 5 here.]

Panel B of Table 5 presents estimates of Eq. (3). The estimates on  $d[t + \tau] \times Treat$  for the prechange period (i.e.,  $\tau = -3$  and -2) are statistically insignificant, pointing toward no differential pre-trends between the two groups. Combined with the evidence that the establishment covariates are similar between the two groups pre-event, this result shows that the treated and control establishments are comparable before the events. The estimates on  $d[t + \tau] \times Treat$  for the post-change period are all negative and statistically and economically significant, indicating that establishments that experience a large increase in block institutional ownership undergo employment and payroll losses relative to plausibly comparable establishments that experience a large increase in overall institutional ownership but not block ownership. The economic magnitude of the estimates is comparable to that of the OLS estimates in Table 3. For example, the estimates for  $d[t+2] \times Treat$  are -0.021 and -0.023 for log employment and payroll, which are similar to the magnitudes of the effect of the same, relative change in block institutional ownership (= (0.110 - (-0.021)) × (-0.150) or (-0.166) = -0.020 or -0.022). Overall, these results corroborate the validity of our baseline estimates and indicate that the results are unlikely to be driven by selection of institutional shareholders into firms.

#### 4. Mechanism

The analysis so far shows consistently negative within-establishment associations between powerful institutional ownership and quantity of labor. This section explores the mechanisms that underlie the baseline relations.

#### 4.1. Bargaining with employees – labor union and labor market concentration

A plausible mechanism for powerful shareholders to affect the firm's labor force is through bargaining. To the extent that the firm and labor jointly create rents that are to be shared, an increase in relative bargaining power of shareholders vis-à-vis employees would lead to smaller employment and payroll, as we show. Following the literature on the bargaining power of employees (e.g., Matsa, 2010; and Benmelech, Bergman, and Kim, 2020), we measure employees' bargaining power relative to shareholders using industry-level unionization rates and industry-local area (three-digit SIC-commuting zone) level measure of labor market concentration measured by the HHI of firm-level employment. We estimate the following regression that interacts large institutional ownership with the two proxies for labor bargaining power, unionization rates and the local labor market HHI, in turn:

 $y_{ijkt} = \alpha_i + \alpha_{kt} + \alpha_{ct} + \beta_1 Large IO_{jt-1} + \beta_2 Large IO_{jt-1} \times Labor Bargaining_{kct-1} + \beta_3 Labor Bargaining_{kct-1} + \beta_4 Overall IO_{jt-1} + \gamma' X_{it-1} + \varepsilon_{ijkt},$ (4)

where *Labor bargaining*<sub>kd-1</sub> is the one-year lagged fraction of workers covered by collective bargaining in industry *k* or the HHI firm employment in industry *k* and commuting zone *c*,  $a_{ct}$ is industry-by-commuting zone fixed effects, and all other variables are defined as in Eq. (1). The coefficient of interest is  $\beta_2$ , which represents the interaction effect of large, powerful institutional ownership with labor bargaining power on establishment-level employment and payroll.

## [Insert Table 6 here.]

Table 6 presents estimation results for Eq. (4) for the interaction with unionization rates (Panel A) and the local labor market HHI (Panel B). Column 1 of Panel A confirms the

baseline result that within-establishment, an increase in the largest institutional ownership is associated with a significant decrease in employment. Importantly, the positive and significant (at the 1% level) coefficient on "Top 1 ownership  $\times$  Union" in the column shows that high union coverage, which increases bargaining power of employees, mitigates the negative effect of powerful institutional owners on employment. The estimates indicate that in industries where the unionization rate is zero, a ten-percentage point increase in the largest institution's ownership is associated with a 4.1% reduction in employment. However, in industries where collective bargaining coverage is one-SD above its mean (= 18.4%), the same magnitude increase in top institutional ownership is related with a much smaller -0.5% reduction in employment. The estimates in column 5 show a very similar magnitude of the mitigating effect of labor unions on payroll. Thus, the results in these columns are consistent with higher collective bargaining coverage improving labor's bargaining power against shareholders, thereby mitigating the negative effect of large, powerful shareholders on employment and payroll. A mutually-non-exclusive explanation is that the (positive) effect of labor unions on employment and payroll is more pronounced in the presence of powerful institutional shareholders.

In addition, estimates in columns 2-3 and 6-7 show that the negative effects of top five institutional shareholders and total blockholder ownership are similarly mitigated when labor unions provide stronger bargaining power to employees. Across the columns, the interaction effects of large shareholders' ownership and unionization rates are significant at the 1% to 5% levels. Columns 4 and 8 show that labor unions might also mitigate the effect of concentrated institutional ownership, although the estimated coefficients on "HHI (IO)  $\times$  Union" is statistically insignificant at a conventional level.

Finally, the estimates in Panel B of Table 6 further corroborate the bargaining power interpretation. The coefficients on the interaction terms between large shareholder ownership and local labor market concentration are negative and generally statistically significant, indicating that powerful institutional shareholders matter more for employment and payroll in more concentrated labor markets, in which firms have greater bargaining power vis-à-vis workers.

#### 4.2. Impact on labor productivity and shareholder value

Next, to further clarify the mechanism, we examine the impact of powerful shareholders on firm-level measures of firm output per worker vs. shareholder value per worker. If employment losses are due to greater monitoring of CEOs and a reduction in their preference for the quiet life, for example, then we would expect to see an improvement in overall firm output and labor efficiency. By contract, bargaining and rent extraction would predict a loss in joint output. To examine these predictions, we use information on firm revenues from the revenue-enhanced LBD and re-estimate the baseline specification in Eq. (1) using revenues per employee as the dependent variable.<sup>13</sup>

## [Insert Table 7 here.]

The results are summarized in Columns 1 to 4 of Table 7. There is a negative relation between labor productivity and all of the four measures of shareholder power. The coefficient estimates are large and statistically significant for all proxies but the ownership HHI. For example, the estimate for "Total block ownership" in column 3 shows that, when a given establishment experiences a ten-percentage point increase in block institutional ownership, its labor productivity declines by 2.5%. The coefficient is significant at a less than 1% level. Further supporting an incongruence of interests between powerful shareholders and workers, when we repeat the analysis using a measure of shareholder value, shareholder returns per employee,<sup>14</sup> the sign of the relation reverses. Specifically, Columns 5 to 8 of Table 7 show a consistently positive relation between shareholder returns and the four measures of shareholder power. Also the relation with shareholder returns is economically significant. For example, the estimate for "Total block ownership" in column 7 shows that, when a given establishment experiences a ten-percentage point increase in block institutional ownership, its shareholder returns increase by 6.3%. The coefficient is significant at a less than 1% level.

### 4.3. Types of institutional shareholders

 <sup>&</sup>lt;sup>13</sup> The revenue-enhanced LBD is a new development by the Census Bureau and is available beginning in 1997.
 <sup>14</sup> Shareholder returns are defined as the logarithmic change in stock market value per employee, which is measured as the ratio of stock market capitalization to number of employees from Compustat

In our final battery of tests to probe the mechanism, we examine what types of institutional investors drive the relation between shareholder power and establishment-level labor outcomes. Specifically, we follow Bushee's (1998) classification of institutional ownership into "quasi-index," "dedicated," and "transient" types and calculate the percentage of shares owned by these types of institutional investors scaled by overall percentage institutional ownership. In addition, we define "activist" institutional investors following Grennan (2019), who bases a classification on whether a given institution has engaged in shareholder activism campaigns. We then calculate the percentage owned by the identified activist institutional investors and scale it by overall institutional ownership. Thus, these measures capture the relative importance of specific types of institutions among all institutional investors' ownership. We employ the measures of specific shareholder ownership in a variant of Eq. (1) in which the control for overall institutional ownership is dropped, given that the overall ownership is already controlled for by scaling the larger shareholder ownership variables by it.

Panel A of Table 8 presents estimation results. Columns 1 and 5 show that an increase in quasi-index ownership share among overall institutional ownership is associated with increases in employment and payroll. A plausible interpretation of the positive relations is that quasi-index funds are more "lenient" monitors relative to other types of institutions, under which firms expand their workforce and payroll. Similarly, an increase in the share of transient institutional ownership is significantly associated with higher payroll (column 6), although its association with employment is insignificant (column 2).

## [Insert Table 8 here.]

In contrast, we find that the association between the relative ownership share of dedicated institutions, which have long-term investment horizons and monitoring incentives (due to their concentrated holdings), and employment or payroll is significantly negative at the 1% level. The coefficients on "Dedicated IO/Overall IO" in columns 3 and 7 show that a ten-percentage point increase in the relative share of dedicated institutional ownership is associated with 0.7% and 0.8% reductions in employment and payroll, respectively.

The largest magnitude of the negative effect on labor comes from activist institutional investors. The coefficients on "Activist IO/Overall IO" in columns 4 (-0.725) and 8 (-0.761)

are significant at the 1% level and approximately ten times greater in magnitude than those on "Dedicated IO/Overall IO." The estimates suggest that a ten-percentage point increase in activist institutional ownership shares among general institutional ownership would result in 7.3% to 7.6% reductions in establishment employment and payroll. This large, negative association between activist shareholders and employment and wage bills is consistent with evidence in Brav, Jiang, and Kim (2015), who focus on hedge fund activism events and manufacturing plants from 1994 to 2008.

Overall, the results in this section are consistent with dedicated (who hold their concentrated ownership over a relatively long horizon) and activist institutions (who engage in shareholder activism campaigns) pushing for a reduction in labor in the interest of shareholder value. The most direct way for activist (and dedicated) investors to affect firms' labor is through intervention and public engagement with management.

Panel B of Table 8 offers direct supporting evidence for this mechanism. We reestimate our baseline specification in Eq. (1) at the firm level using an indicator for whether a firm becomes the target of a shareholder activism campaign in a given year as the dependent variable. The coefficient estimates are all positive and statistically significant across our four proxies, indicating that powerful shareholder-owned firms are more likely to be the targets of activism campaigns. In addition, textual analysis of the announcements indicates that activism works through two main types of actions, either formal proxy fights and shareholder proposals at annual meetings or soft engagement via public announcements in the press and newswires. Approximately three quarters of the target firms are involved in the former type of campaign, while about two thirds are involved in the latter. As for the stated objectives of the campaigns, for about 45% of the targets they involve firm operations, for about 23% of the targets they involve shareholder rights and maximizing shareholder value, and for about 33% they involve mergers and restructurings.

Anecdotal evidence also suggests that activist campaigns often directly target job cuts as cost cutting measures to reduce operating expenses. For example, see "Canadian Pacific To Cut 4,500 Jobs," *The Wall Street Journal*, December 12, 2012. After activist investor Bill Ackman won a proxy battle at Canadian Pacific, the railroad company decided to cut 23% of their workforce under its new chief executive Hunter Harrison who was appointed by Ackman. Similarly, on November 28, 2012, Icahn Capital LP announced its intention to discuss the operations and direction of Enzon Pharmaceuticals Inc. and the Company's plans to manage expenses and preserve value for shareholders. And on April 26, 2001, GAMCO Investors, Inc. announced that Mario Gabelli sent a letter to the Board of Directors of Carter-Wallace, Inc. expressing its views that restructuring of the business from an operational and financial point of view was needed.

### 5. Conclusion

Labor markets in the U.S. have experienced a pronounced transformation over the last decades, with labor income stagnating despite strong productivity and profit growth. This paper sets forth an explanation based on the ownership structure of public firms. Motivated by the classical theory of the firm based on conflicts of interests between shareholders and stakeholders, we conjecture that employment and labor income should be lower at firms owned by more influential and highly concentrated institutional shareholders, because the objectives and interests of the shareholders more likely conflict with those of the workers. Using confidential micro-level data from the U.S. Census Bureau's Longitudinal Business Database from 1982-2015, we provide comprehensive evidence that supports this explanation, namely that establishments of firms owned by large and highly concentrated institutional shareholders have considerably lower employment and payroll.

As a first step toward examining the aggregate implications, we perform a simple counterfactual exercise to gauge how much of the time-series decline in labor income we can explain with rising shareholder power. Using the main estimates from Table 3 (Panel A, Column 3), we calculate the implied time-series effect given the change in average block ownership since 1980. The results shown in Figure 2 indicate that the rise in shareholder power explains about a 2 percentage-point decline, which is roughly 1/3 of the overall decline in total wages and salaries relative to GDI. Overall, these findings suggest that there is scope for employment policies aimed at reforming shareholder capitalism to broaden the objective of the firm beyond the narrow maximization of shareholder value, such as increasing labor representation on boards or allowing for labor participation in firm governance.



Figure 2: Implications for the Trend in Employee Compensation-to-GDI Ratio in the US, 1980–2014. This figure plots the ratio of total wage and salary accruals to domestic gross income (GDI) (blue solid line) and the counterfactual ratio of total wage and salary accruals to domestic gross income (GDI) (red dotted line) in the US for the period 1980–2014. The counterfactual ratio is constructed as the product of the estimate for block institutional ownership from Table 3 (Panel A, Column 3) times the change in average blockholder ownership in any given year relative to 1980. The compensation-to-GDI ratio is from the Federal Reserve Economic Data. Block institutional ownership is the percentage owned by institutional blockholders, defined as the institutional investors with more than 5% holdings as filed through 13D, 13F, or 13G filings. The average block ownership is calculated excluding firms in the agriculture, financial, utility, and public administration sectors.

### References

- Acemoglu, Daron, 2002, "Technical Change, Inequality, And the Labor Market," Journal of Economic Literature, 40(1), 7-72.
- Acemoglu, Daron, and David H. Autor, 2011, "Skills, Tasks and Technologies: Implications for Employment and Earnings," in Handbook of Labor Economics, vol. 4B, David Card and Orley Ashenfelter, eds. (Amsterdam: Elsevier, 2011), 1043–71.
- Acemoglu, Daron, David Autor, David Dorn, Gordon H. Hanson, and Brendan Price, 2016, "Import Competition and the Great US Employment Sag of the 2000s," Journal of Labor Economics, 34(1), 141 – 98.
- Autor, David H., David Dorn, and Gordon H. Hanson, 2013, "The China Syndrome: Local Labor Market Effects of Import Competition in the United States," American Economic Review, vol. 103(6), 2121-68.
- Autor, David, David Dorn, Lawrence F. Katz, Christina Patterson, and John Van Reenen, 2020, "The Fall of the Labor Share and the Rise of Superstar Firms," Quarterly Journal of Economics, Forthcoming.
- Benmelech, Efraim, Nittai Bergman, and Hyunseob Kim, 2020, "Strong Employers and Weak Employees: How Does Employer Concentration Affect Wages?," Journal of Human Resources, Forthcoming.
- Bertrand, Marianne and Sendhil Mullainathan, 2003, "Enjoying the Quiet Life? Managerial Behavior Following Anti-Takeover Legislation," Journal of Political Economy, 11, 1043-75.
- Brav, Alon, Wei Jiang, and Hyunseob Kim, 2015, "The Real Effects of Hedge Fund Activism: Productivity, Asset Allocation, and Labor Outcomes," Review of Financial Studies 28, 2723-69.
- Bushee, Brian J., 1998, "The Influence of Institutional Investors on Myopic R&D Behavior," Accounting Review, 73(3), 305-33.
- Card, David, Jorg Heining, and Patrick Kline, 2013, "Workplace Heterogeneity and the Rise of West German Wage Inequality," The Quarterly Journal of Economics, 128(3), 967– 1015.

- Cronqvist, Henrik, Fredrik Heyman, Mattias Nilsson, Helena Svaleryd, and Jonas Vlachos, 2009, "Do Entrenched Managers Pay Their Workers More?," Journal of Finance, 64(1), 309-39.
- Davis, Steven, John Haltiwanger, and Scott Schuh, 1996, Job Creation and Destruction (MIT Press, Cambridge, MA).
- Davis, Steven, John Haltiwanger, Kyle Handley, Ron Jarmin, Josh Lerner, and Javier Miranda, 2014, "Private Equity, Jobs, and Productivity, American Economic Review," 104(12), 3956-90.
- Fort, Teresa C., and Shawn Klimek. 2018. "The Effects of Industry Classification Changes on US Employment Composition." CES Working Paper 18-28.
- Grennan, Jillian, 2019, "A Corporate Culture Channel: How Increased Shareholder Governance Reduces Firm Value," Working paper, Duke University.
- Jarmin, Ron, and Javier Miranda, 2002, "The Longitudinal Business Database," CES Working Paper 02-17.
- Jensen, Michael, and William Meckling, 1976, "Theory of the Firm: Managerial Behavior, Agency Costs and Capital Structure," Journal of Financial Economics, 3, 11-25.
- Krugman, Paul, 2015, "Challenging the Oligarchy," The New York Review of Books (December).
- Lichtenberg, Frank R., and Donald Siegel, 1990, "The effects of leveraged buyouts on productivity and related aspects of firm behavior," Journal of Financial Economics, 27(1), 165-94.
- Moretti, Enrico, 2011, "Local Labor Markets." In Handbook of Labor Economics 4B, eds. David Card, and Orley Ashenfelter, 1237–313. Amsterdam: Elsevier.
- Pierce, Justin R., and Peter K. Schott. 2016. "The Surprisingly Swift Decline of US Manufacturing Employment." American Economic Review, 106 (7): 1632-62.
- Posner, Eric, 2019, "Milton Friedman Was Wrong," The Atlantic (August).
- Shleifer, Andrei, and Robert W. Vishny, 1986, "Large Shareholders and Corporate Control," Journal of Political Economy 94(3),461-88.

- Shleifer, Andrei and Lawrence H. Summers. "Breach of Trust in Hostile Takeovers," From Corporate Takeovers: Causes and Consequences, edited by Alan J. Auerbach, pp. 33-56. Chicago: The University of Chicago Press, 1988.
- Shleifer, Andrei, and Robert W. Vishny, 1997, "A Survey of Corporate Governance," Journal of Finance 52 (2): 737-83.
- Schoar, Antoinette, 2002, "Effects of Corporate Diversification on Productivity," Journal of Finance 57(6): 2379–403.
- Song, Jae, David J Price, Fatih Guvenen, Nicholas Bloom, and Till von Wachter, 2019, "Firming Up Inequality," The Quarterly Journal of Economics, 134(1), 1-50.
- Stein, J., 2003, "Agency, Information and Corporate Investment," in G.M. Constantinides, M. Harris and R. Stulz, eds.: Handbook of the Economics of Finance (Elsevier, Amsterdam).
- Stiglitz, Joseph E., 2019, People, Power, and Profits: Progressive Capitalism for an Age of Discontent, W. W. Norton & Company.

Table 1: Summary statistics on establishment observations from the LBD, 1982–2015

Panel A:	All	indus	tries
----------	-----	-------	-------

	Mean	STD
Overall IO	0.597	0.266
Top 1 ownership	0.087	0.065
Top 5 ownership	0.237	0.120
Total block ownership	0.147	0.145
HHI of institutional ownership	0.072	0.110
Total employment	61.98	294.60
Log total employment	2.781	1.492
Total payroll (\$000)	2,471.0	70,880.0
Log total payroll	5.811	1.715
Average wage (\$000)	29.530	28.410
Log average wage	3.030	0.847
Establishments per firm	2,084	2,531
Establishments per segments (SIC3)	1,546	2,169
Establishments per segments (SIC4)	1,483	2,107
Establishment age (/100)	0.108	0.084
Emp growth rate	-0.004	0.356
Pay growth rate	0.005	0.422
Quasi index IO	0.585	0.224
Dedicated IO	0.131	0.164
Transient IO	0.215	0.152
Activist IO	0.005	0.017
Union	0.080	0.104
Observations	7,340,000	-

# Panel B: Manufacturing

	Mean	STD
Overall IO	0.537	0.244
Top 1 ownership	0.081	0.059
Top 5 ownership	0.215	0.111
Total block ownership	0.120	0.128
HHI of institutional ownership	0.077	0.114
Total employment	295.50	792.10
Log total employment	4.517	1.645
Total payroll (\$000)	13,670.0	64,110.0
Log total payroll	8.072	1.755
Average wage (\$000)	42.680	29.310
Log average wage	3.549	0.660
Establishments per firm	331.9	829.3
Establishments per segments (SIC3)	19.4	34.9
Establishments per segments (SIC4)	16.7	33.8
Establishment age (/100)	0.149	0.096

Emp growth rate	-0.017	0.316
Pay growth rate	0.006	0.437
Quasi index IO	0.597	0.209
Dedicated IO	0.158	0.170
Transient IO	0.193	0.141
Activist IO	0.005	0.018
Unionization rate	0.185	0.117
Observations	533,000	-

=

Note: This table presents descriptive statistics on establishment-year observations used in the analysis from the Longitudinal Business Database (LBD) for the period 1982-2015. We require each observation in the sample to have the lagged observation. "Total wage" is the total payroll in thousand dollars; "Total employment" is the number of total employees; "Average wage" is computed as total wage divided by total employment (in thousand dollars); "Establishments per segment" is the number of establishments in a given three-digit SIC industry segment of a given firm; "Establishments per firm" is the total number of establishments of a given firm; "Establishments age" is the number of years since a plant's birth, which is proxied either by the flag for establishments birth in the LBD or by its first appearance in the LBD, whichever is earliest; "Emp (pay) growth rate" is the change in employment (payroll) scaled by sum of current and lagged employment (payroll); and "Unionization rate" is the industry-level percentage of the workforce covered by collective bargaining collected from the CPS. The institutional holdings data are obtained from Thomson-Reuters 13F SEC filings. The ownership variables are defined as the percentage of shares outstanding owned by a given type of institutional investors: "Overall IO" is the percentage owned by all institutional investors, "Top 1 ownership," "Top 5 ownership," and "Top 10 ownership," are the percentage owned by the largest, the largest 5, and the largest 10 institutional investors, respectively. "Top block ownership" is the percentage owned by blockholders, which are defined as the institutional investors with more than 5% holdings as filed through 13D, 13F, or 13G filings. "Quasi index IO," "Dedicated IO,' and "Transient IO" are the percentage owned by quasi-index, dedicated, and transient institutional investors as classified by Bushee (1998), scaled by overall institutional ownership. "Activist IO" is the percentage owned by activist institutional investors as classified by Grennan (2019) based on institutions who have engaged in shareholder activism campaigns, scaled by overall institutional ownership. "HHI of ownership" is the ownership concentration Herfindahl-Hirschman Index, which approaches zero when the firm is owned by a large number of institutional investors of relatively equal size and reaches its maximum of one when it is controlled by a single large institutional investor. The number of observations is rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

	(1)	(2)	(3)	(4)
Dep. Var.:	Log	emp.	Log	; pay
Top 1 ownership	-0.217	-	-0.205	-
	-4.55	-	-5.22	-
Top 5 ownership	-	-0.233	-	-0.253
	-	-6.00	-	-6.48
Overall IO	0.069	0.123	0.084	0.149
	2.75	4.85	3.35	5.84
log(est's per segment)	-0.004	-0.004	0.003	0.003
	-0.76	-0.81	0.56	0.50
log(est's per firm)	0.028	0.026	0.032	0.029
	2.84	2.62	2.59	2.40
Establishment fixed effects	Υ	Y	Y	Υ
Industry-year fixed effects	Υ	Y	Y	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.9170	0.9170	0.9233	0.9233

Table 2: Effect of top institutional owners on establishment-level employment and payroll

#### Panel A: Full sample

### Panel B: Manufacturing

	(1)	(2)	(3)	(4)
Dep. Var.:	Log	emp.	Log	pay
Top 1 ownership	-0.295	-	-0.320	-
	-5.13	-	-5.35	-
Top 5 ownership	-	-0.290	-	-0.322
	-	-6.76	-	-7.39
Overall IO	0.194	0.261	0.228	0.304
	8.78	9.73	9.43	10.55
log(est's per segment)	0.005	0.004	0.007	0.006
	0.96	0.84	1.26	1.12
log(est's per firm)	0.022	0.019	0.018	0.014
	5.19	4.44	3.72	3.02
Establishment fixed effects	Υ	Y	Y	Υ
Industry-year fixed effects	Υ	Y	Y	Υ
Observations	533,000	533,000	533,000	533,000
R <sup>2</sup>	0.9260	0.9260	0.9231	0.9231

Note: This table presents the effect of institutional ownership on employment and payroll of establishments. The dependent variable is the log of employment (Columns 1 and 2) and payroll (Columns 3 and 4) as defined in Table 1. Variables for institutional ownership are lagged by one year. Panel A (Panel B) presents estimates using the full (manufacturing) sample. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

### Table 3: Effect of blockholders and ownership concentration on establishment-level employment and payroll

	(1)	(2)	(3)	(4)
Dep. Var.:	Log	emp.	Log	pay
Total block ownership	-0.150	-	-0.166	-
	-5.14	-	-5.69	-
HHI (institutional ownership)	-	-0.103	-	-0.085
	-	-4.55	-	-2.78
Overall IO	0.086	0.016	0.110	0.035
	3.69	0.68	4.28	1.38
log(est's per segment)	-0.004	-0.004	0.003	0.003
	-0.76	-0.70	0.55	0.60
log(est's per firm)	0.027	0.029	0.030	0.033
	2.72	2.92	2.47	2.67
Establishment fixed effects	Υ	Υ	Υ	Υ
Industry-year fixed effects	Y	Υ	Υ	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.9170	0.9170	0.9233	0.9233

#### Panel A: Full sample

#### Panel B: Manufacturing

	(1)	(2)	(3)	(4)
Dep. Var.:	Log	emp.	Log	pay
Total block ownership	-0.171	-	-0.195	-
	-6.37	-	-6.83	-
HHI (institutional ownership)	-	-0.157	-	-0.173
	-	-7.02	-	-7.20
Overall IO	0.215	0.129	0.255	0.157
	9.82	6.42	10.57	7.43
log(est's per segment)	0.005	0.005	0.006	0.007
	0.92	1.02	1.21	1.32
log(est's per firm)	0.020	0.022	0.016	0.018
	4.82	5.36	3.33	3.83
Establishment fixed effects	Υ	Υ	Υ	Υ
Industry-year fixed effects	Υ	Υ	Υ	Υ
Observations	533,000	533,000	533,000	533,000
R <sup>2</sup>	0.9260	0.9260	0.9231	0.9231

Note: This table presents the effect of blockholder ownership and concentration of institutional ownership on employment and payroll of establishments. The dependent variable is the log of employment (Columns 1 and 2) and payroll (Columns 3 and 4) as defined in Table 1. Variables for institutional ownership are lagged by one year. Panel A (Panel B) presents estimates using the full (manufacturing) sample. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

### Table 4: Robustness of baseline results

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:		Log	emp.			Log	pay	
Top 1 ownership	-0.286	-	-	-	-0.274	-	-	-
	-4.98	-	-	-	-5.06	-	-	-
Top 5 ownership	-	-0.256	-	-	-	-0.285	-	-
	-	-5.95	-	-	-	-6.98	-	-
Total block IO	-	-	-0.156	-	-	-	-0.173	-
	-	-	-5.45	-	-	-	-6.27	-
HHI (IO)	-	-	-	-0.108	-	-	-	-0.091
	-	-	-	-4.50	-	-	-	-3.43
Overall IO	0.072	0.126	0.086	0.013	0.087	0.154	0.110	0.032
	3.21	5.31	4.03	0.63	4.04	6.61	4.84	1.41
Establishment-level controls	Y	Υ	Υ	Y	Y	Y	Y	Y
Establishment fixed effects	Υ	Υ	Υ	Y	Υ	Y	Υ	Υ
Industry-year fixed effects	Υ	Υ	Y	Y	Υ	Υ	Υ	Υ
County-year fixed effects	Y	Υ	Υ	Y	Y	Y	Y	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.9186	0.9186	0.9186	0.9186	0.9248	0.9248	0.9248	0.9247

Panel A.1: Controlling for time-varying local labor market conditions

## Panel A.2: Controlling for time-varying local industry-labor market conditions

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:		Log	emp.			Log	; pay	
Top 1 ownership	-0.313	-	-	-	-0.320	-	-	-
	-4.64	-	-	-	-5.40	-	-	-
Top 5 ownership	-	-0.269	-	-	-	-0.315	-	-
	-	-5.48	-	-	-	-7.09	-	-
Total block IO	-	-	-0.165	-	-	-	-0.196	-
	-	-	-5.13	-	-	-	-6.70	-

HHI (IO)	-	-	-	-0.093	-	-	-	-0.086
	-	-	-	-3.32	-	-	-	-3.85
Overall IO	0.083	0.137	0.096	0.020	0.098	0.169	0.122	0.035
	3.17	5.12	4.05	0.86	5.15	7.77	6.09	1.64
Establishment-level controls	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ
Establishment fixed effects	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ
Industry-CZ-year fixed effects	Υ	Y	Υ	Υ	Υ	Υ	Υ	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
<u>R<sup>2</sup></u>	0.935	0.935	0.935	0.935	0.934	0.934	0.934	0.934

Panel B: Moving average of institutional ownership measures

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:		Log	emp.			Log	pay	
Top 1 IO (MA)	-0.335	-	-	-	-0.303	-	-	-
	-4.34	-	-	-	-4.04	-	-	-
Top 5 IO (MA)	-	-0.321	-	-	-	-0.348	-	-
	-	-5.82	-	-	-	-6.34	-	-
Total block IO (MA)	-	-	-0.202	-	-	-	-0.221	-
	-	-	-5.55	-	-	-	-5.78	-
HHI (IO) (MA)	-	-	-	-0.120	-	-	-	-0.086
	-	-	-	-3.88	-	-	-	-2.32
Overall IO (MA)	0.089	0.160	0.114	0.022	0.099	0.185	0.137	0.042
	2.97	5.02	3.82	0.75	3.29	5.85	4.22	1.25
Establishment-level controls	Y	Y	Υ	Υ	Υ	Υ	Y	Υ
Establishment fixed effects	Y	Y	Y	Υ	Υ	Υ	Υ	Y
Industry-year fixed effects	Υ	Y	Y	Υ	Υ	Υ	Υ	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.9170	0.9170	0.9170	0.9170	0.9233	0.9233	0.9233	0.9233

Panel C: Top and block ownership scaled by overall institutional ownership

	(1)	(2)	(3)	(4)	(5)	(6)
Dep. Var.:		Log emp.			Log pay	
Top 1 IO/overall IO	-0.110	-	-	-0.096	-	-
	-4.81	-	-	-3.43	-	-
Top 5 IO/overall IO	-	-0.134	-	-	-0.142	-
	-	-6.59	-	-	-6.46	-
Total block IO/overall IO	-	-	-0.056	-	-	-0.052
	-	-	-3.71	-	-	-3.17
Establishment-level controls	Υ	Y	Y	Y	Y	Υ
Establishment fixed effects	Y	Υ	Υ	Υ	Υ	Υ
Industry-year fixed effects	Y	Y	Y	Y	Y	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.9170	0.9170	0.9170	0.9233	0.9233	0.9233

### Panel D: Not controlling for Overall IO

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:		Log	emp.			Log	pay	
Top 1 ownership	-0.181	-	-	-	-0.146	-	-	-
	-3.2	-	-	-	-2.29	-	-	-
Top 5 ownership	-	-0.095	-	-	-	-0.088	-	-
	-	-2.46	-	-	-	-2.02	-	-
Total block IO	-	-	-0.081	-	-	-	-0.079	-
	-	-	-2.84	-	-	-	-2.57	-
HHI (IO)	-	-	-	-0.090	-	-	-	-0.075
	-	-	-	-4.47	-	-	-	-2.55
Establishment-level controls	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ
Establishment fixed effects	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Y
Industry-year fixed effects	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Y
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.917	0.917	0.917	0.917	0.9233	0.9233	0.9233	0.9233

Note: This table examines the robustness of the baseline effect of institutional ownership on employment and payroll of establishments. In Panels A and B, the dependent variable is the log of employment (Columns 1 through 4) and the log of payroll (Columns 5 through 8) as defined in Table 1. In Panel C, the

dependent variable is the log of employment (Columns 1 through 3) and the log of payroll (Columns 4 through 6). Panel A.1 includes county-by-year fixed effects, Panel A.2 includes county-by-commuting zone-by-year fixed effects, Panel B uses the two-year moving average of institutional ownership measures, and Panel C scales top and block institutional ownership measures by overall institutional ownership, as defined in Table 1. In Panel D, we do not include the Overall IO variable as control. Variables for institutional ownership are lagged by one year. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

# Table 5: Identification – Difference-in-differences analysis

	(1)	(2)	(3)	(4)	(5)
	Trea	ated	Con	trol	(1) - (3)
	Mean	STD	Mean	STD	<i>t</i> -statistic
Total block ownership <sub>t-1</sub>	0.000	0.000	0.147	0.122	-24.44
Overall institutional ownership <sub>t-1</sub>	0.443	0.219	0.529	0.222	-5.17
Change in total block ownershipt-1	0.110	0.096	-0.021	0.051	21.04
Change in overall institutional ownershipt-1	0.093	0.161	0.113	0.083	-1.74
Log total employment <sub>t-1</sub>	2.816	1.492	2.771	1.463	0.71
Log total wages <sub>t-1</sub>	5.708	1.711	5.673	1.687	0.49
Log average wage <sub>t-1</sub>	2.892	0.833	2.902	0.810	-0.20
Establishments per firm <sub>t-1</sub>	5.662	2.218	5.738	2.104	-0.44
Establishments per segments <sub>t-1</sub> (SIC3)	6.504	1.734	6.480	1.651	0.17
Establishment age <sub>t-1</sub> (/100)	0.087	0.070	0.088	0.075	-0.45
Union <sub>t-1</sub>	0.088	0.106	0.082	0.108	0.80
Observations	470,000	-	830,000	-	-

## Panel A: Descriptive statistics and diagnostic tests

# Panel B: Difference-in-differences estimates

	(1)	(2)
Dep. Var.:	Log emp.	Log pay
Treat $\times$ d[t-3]	-0.006	0.003
	-0.60	0.39
Treat $\times$ d[t-2]	-0.005	-0.001
	-0.69	-0.12
Treat $\times$ d[t-1]	0.000	0.000
	-	-
$Treat \times d[t_0]$	-0.009	-0.011
	-1.32	-1.44
$Treat \times d[t+1]$	-0.009	-0.023
	-0.98	-2.66
$Treat \times d[t+2]$	-0.021	-0.023
	-2.21	-2.76
Treat $\times$ d[t+3]	-0.014	-0.022
	-1.34	-2.29
$Treat \times d[t+4]$	-0.021	-0.021
	-1.86	-1.91
log(est's per segment)	-0.009	-0.005
	-1.82	-1.02
log(est's per firm)	0.028	0.027
	4.63	4.19
Standalone indicators for d[t-3]		
through d[t+4]	Y	Y
Establishment fixed effects	Y	Υ
Industry-year fixed effects	Y	Y
Observations	8,100,000	8,100,000

	R <sup>2</sup> 0.92 <sup>4</sup>	41 0.9362
--	----------------------------------	-----------

Note: This table repeats the baseline analysis of the effect of institutional ownership on employment and payroll of establishments in a difference-in-differences (DD) setting. Panel A presents descriptive statistic for the sample employed and results of diagnostic tests. Panel B presents the main DD estimates. The dependent variable is the log of employment (Column 1) and payroll (Column2) as defined in Table 1. The sample is limited to observations that involve more than 5% increases in block ("Treat") and overall institutional ownership ("Control"). The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

Panel A: Interaction with Unionized Labor									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Dep. Var.:		Log	emp.			Log	; pay		
Top 1 ownership	-0.414	-	-	-	-0.398	-	-	-	
	-5.00	-	-	-	-4.77	-	-	-	
Top 1 ownership × Union	1.843	-	-	-	1.826	-	-	-	
	3.2	-	-	-	2.46	-	-	-	
Top 5 ownership	-	-0.330	-	-	-	-0.346	-	-	
	-	-5.97	-	-	-	-5.9	-	-	
Top 5 ownership × Union	-	1.107	-	-	-	0.939	-	-	
	-	3.07	-	-	-	2.08	-	-	
Total block ownership	-	-	-0.202	-	-	-	-0.224	-	
	-	-	-5.24	-	-	-	-5.66	-	
Total block ownership $\times$ Union	-	-	0.819	-	-	-	0.886	-	
	-	-	2.86	-	-	-	2.56	-	
HHI (IO)	-	-	-	-0.111	-	-	-	-0.116	
	-	-	-	-3.06	-	-	-	-4.09	
HHI (IO) $\times$ Union	-	-	-	0.093	-	-	-	0.350	
	-	-	-	0.37	-	-	-	0.94	
Overall IO	0.073	0.125	0.085	0.016	0.088	0.153	0.109	0.036	
	3.13	5.10	3.73	0.68	3.70	5.97	4.29	1.40	
Establishment-level controls	Υ	Υ	Υ	Υ	Υ	Υ	Y	Υ	
Establishment fixed effects	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	
Industry-year fixed effects	Y	Y	Υ	Υ	Υ	Y	Y	Υ	
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	
R <sup>2</sup>	0.9170	0.9170	0.9170	0.9170	0.9233	0.9233	0.9233	0.9233	

## Table 6: Mechanism – Interaction with Unionized Labor and Labor Market Concentration

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Dep. Var.:		Log	emp.			Log pay			
Top 1 ownership	-0.178	-	-	-	-0.157	-	-	-	
	-2.34	-	-	-	-1.68	-	-	-	
Top 1 ownership × HHI (SIC3-CZ)	-1.507	-	-	-	-1.804	-	-	-	
	-2.00	-	-	-	-1.78	-	-	-	
Top 5 ownership	-	-0.202	-	-	-	-0.200	-	-	
	-	-3.46	-	-	-	-3.37	-	-	
Top 5 ownership × HHI (SIC3-CZ)	-	-0.747	-	-	-	-1.280	-	-	
	-	-1.39	-	-	-	-1.93	-	-	
Total block ownership	-	-	-0.105	-	-	-	-0.108	-	
-	-	-	-2.38	-	-	-	-2.29	-	
Total block ownership × HHI (SIC3-CZ)	-	-	-0.693	-	-	-	-1.003	-	
	-	-	-1.59	-	-	-	-1.81	-	
HHI (IO)	-	-	-	-0.048	-	-	-	-0.080	
	-	-	-	-1.99	-	-	-	-3.84	
HHI (IO) $\times$ HHI (SIC3-CZ)	-	-	-	-0.485	-	-	-	-0.065	
	-	-	-	-2.46	-	-	-	-0.23	
Overall IO	0.079	0.135	0.093	0.018	0.093	0.165	0.117	0.035	
	3.14	5.24	4.04	0.81	4.68	7.52	5.75	1.69	
Establishment-level controls	Υ	Y	Y	Υ	Y	Υ	Y	Υ	
Establishment fixed effects	Y	Y	Y	Y	Y	Υ	Y	Υ	
Industry-CZ-year fixed effects	Y	Y	Y	Y	Y	Υ	Υ	Υ	
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	
$\mathbb{R}^2$	0.9350	0.9350	0.9350	0.9350	0.9393	0.9394	0.9394	0.9393	

#### Panel B: Interaction with Labor Market Concentration

Note: This table examines unionized labor as a mechanism behind the effect of institutional ownership on employment and payroll of establishments. Panel A reports results for the interaction with unionization. Panel B reports results for the interaction with labor market concentration. The dependent variable is the log of employment (Columns 1 through 4) and the log of payroll (Columns 5 through 8) as defined in Table 1. Variables for institutional ownership and industry-level unionization rates ("Union") are lagged by one year. The standalone "Union" variable (Panel A) and local labor market HHI at the three-digit SIC-by-commuting zone level (Panel B) are included in all regressions but omitted in the table. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Dep. Var.:		Log labor f	productivity		Shareholder Value per Employee				
Top 1 ownership	-0.422	-	-	-	0.744	-	-	-	
	-4.10	-	-	-	14.13	-	-	-	
Top 5 ownership	-	-0.379	-	-	-	0.702	-	-	
	-	-5.61	-	-	-	18.80	-	-	
Total block IO	-	-	-0.247	-	-	-	0.626	-	
	-	-	-5.53	-	-	-	23.87	-	
HHI of block ownership	-	-	-	-0.013	-	-	-	0.244	
	-	-	-	-0.40	-	-	-	13.44	
Overall IO	0.154	0.245	0.199	0.093	-0.708	-0.882	-0.855	-0.538	
	4.27	5.88	5.32	2.80	-34.02	-35.60	-36.88	-28.57	
log(est's per firm)	-0.103	-0.104	-0.104	-0.101	-	-	-	-	
	-9.04	-9.16	-9.16	-8.84	-	-	-	-	
Firm fixed effects	Υ	Υ	Υ	Υ	Y	Υ	Y	Υ	
Industry-year fixed effects	Υ	Υ	Υ	Υ	Y	Υ	Y	Υ	
Observations	45,000	45,000	45,000	45,000	75,474	75,474	75,474	75,474	
R <sup>2</sup>	0.8292	0.8293	0.8293	0.8290	0.3650	0.3662	0.3682	0.3658	

Table 7: Mechanism - Impact on Labor Productivity and Shareholder Value

Note: This table presents the effect of top institutional owners on labor productivity and shareholder value. The dependent variable is the log of labor productivity, which is measured as revenues per employee from the revenue-enhanced LBD (Columns 1 through 4) and the logarithmic change in stock market value per employee, which is measured as the ratio of stock market capitalization to number of employees from Compustat (Columns 5 through 8). Variables for institutional ownership are lagged by one year. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations in Columns 1 to 4 are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:		Log	emp.			Log	pay	
Quasi index IO	0.051	-	-	-	0.044	-	-	-
	3.37	-	-	-	3.06	-	-	-
Transient IO	-	0.009		-	-	0.034		-
	-	0.51		-	-	1.99		-
Dedicated IO	-		-0.072	-	-		-0.075	-
	-		-4.21	-	-		-4.23	-
Activist IO	-	-	-	-0.725	-	-	-	-0.761
	-	-	-	-3.28	-	-	-	-3.84
Overall IO	0.035	0.028	0.037	0.034	0.052	0.045	0.055	0.051
	1.56	1.18	1.65	1.47	2.11	1.77	2.27	2.03
Establishment-level controls	Υ	Y	Y	Y	Y	Y	Υ	Υ
Establishment fixed effects	Υ	Y	Y	Υ	Y	Y	Υ	Υ
Industry-year fixed effects	Y	Y	Y	Y	Υ	Y	Y	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.9170	0.9170	0.9170	0.9170	0.9233	0.9233	0.9233	0.9233

Table 8: Mechanism - Variation by institutional investor type and impact on shareholder activism

Panel A: Variation by institutional investor type

Note: This table presents the effect of institutional ownership on employment and payroll of establishments across different types of institutions. The dependent variable is the log of employment (Columns 1 through 4) and the log of payroll (Columns 5 through 8) as defined in Table 1. Variables for institutional ownership are lagged by one year. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.

I AUCI D. IIIIDACI UII ACIIVISII	Panel	<b>B</b> :	Imp	act o	on	activism
----------------------------------	-------	------------	-----	-------	----	----------

	(1)	(2)	(3)	(4)
Dep. Var.:		1(Activ	vism)	
Top 1 ownership	0.543	-	-	-
	1.95	-	-	-
Top 5 ownership	-	0.679	-	-
	-	3.53	-	-
Total block IO	-	-	0.619	-
	-	-	3.42	-
HHI of block ownership	-	-	-	0.037
	-	-	-	1.44
Overall IO	0.012	-0.179	-0.250	0.092
	0.17	-1.88	-2.14	1.47
Firm fixed effects	Y	Y	Y	Y
Industry-year fixed effects	Y	Y	Y	Y
Observations	38,444	38,444	38,444	38,444
R <sup>2</sup>	0.2343	0.2346	0.2690	0.2342

Note: This table presents the effect of top institutional owners on shareholder activism. The dependent variable is an indicator that equals one for years when a firm becomes the target of an announced shareholder activism campaign. Variables for institutional ownership are lagged by one year. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. Data on announcements of shareholder activism campaigns is from Capital IQ's Key Developments database for 2000-2010.

### Appendix Table A1: Alternative specifications - growth rates of employment and payroll

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:		Growth r	ate (emp.)			Growth	rate (pay)	
Top 1 IO	-0.030	-	-	-	-0.080	-	-	-
	-1.96	-	-	-	-3.05	-	-	-
Top 5 IO	-	-0.036	-	-	-	-0.079	-	-
	-	-3.13	-	-	-	-4.81	-	-
Total block IO	-	-	-0.026	-	-	-	-0.061	-
	-	-	-3.01	-	-	-	-5.07	-
HHI of block ownership	-	-	-	0.002	-	-	-	-0.025
	-	-	-	0.19	-	-	-	-1.39
Overall IO	0.004	0.012	0.009	0.001	0.019	0.036	0.029	0.006
	0.79	2.04	1.54	0.21	3.08	4.67	4.16	1.02
Establishment-level controls	Υ	Υ	Υ	Y	Υ	Υ	Y	Υ
Industry-year fixed effects	Υ	Υ	Y	Y	Υ	Y	Y	Υ
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
R <sup>2</sup>	0.0392	0.0392	0.0392	0.0392	0.3183	0.3184	0.3184	0.3183

#### Panel A: Employment and payroll growth

#### Panel B: Weighted by establishment employment

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.:	Growth rate (emp.)				Growth rate (pay)			
Top 1 IO	-0.006	-	-	-	-0.068	-	-	-
	-0.29	-	-	-	-2.54	-	-	-
Top 5 IO	-	-0.017	-	-	-	-0.073	-	-
	-	-1.35	-	-	-	-4.50	-	-
Total block IO	-	-	-0.013	-	-	-	-0.055	-
	-	-	-1.59	-	-	-	-4.83	-
HHI of block ownership	-	-	-	0.011	-	-	-	-0.025
	-	-	-	0.75	-	-	-	-1.49
Overall IO	-0.005	0.000	-0.002	-0.004	0.017	0.033	0.026	0.005
	-0.83	-0.02	-0.25	-0.81	3.43	5.36	4.91	0.78
Establishment-level controls	Υ	Υ	Υ	Υ	Υ	Υ	Y	Υ
Industry-year fixed effects	Υ	Υ	Υ	Υ	Υ	Υ	Y	Y
Observations	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000	7,340,000
<u>R<sup>2</sup></u>	0.0418	0.0418	0.0418	0.0418	0.3644	0.3645	0.3645	0.3644

Note: This table examines the effect of institutional ownership on employment and payroll growth rates of establishments. The dependent variable is the symmetric growth rate of employment (Columns 1 through 4) and payroll (Columns 5 through 8) as defined in Table 1. Panel B estimates weighted regressions using the log employment of establishments as the weight. Variables for institutional ownership are lagged by one year. The *t*-statistics based on standard errors adjusted for sample clustering at the firm are reported below the coefficient estimates. The numbers of observations are rounded to the nearest thousand to follow the Census Bureau's disclosure rules.